



To the Editor of World Journal of Gastroenterology

Murcia, 23rd June 2017

Dear Editor,

I am pleased to submit the revised version of the article N° 34103 entitled “**Evidences supporting the vascular etiology of post-double balloon enteroscopy pancreatitis. Study in porcine mode**”.

All the changes highlighted in the pre-reviewed version have been considered and opportunely commented. A change to the format of Systematic Review is mentioned, however as explained in previous emails addressed to you, there is something misunderstood here which made us to produce a traditional Review.

The paper was agreed after invitation from Mr. Lian-Sheng Ma in 21st June 2016. We took for granted we agreed to produce a “Review” article, while did not realize it was registered as “Systematic Review”. Adapting the manuscript to a Systematic Review format will completely change the focus and aims of the manuscript, so with your expected understanding, we are keeping it as a Review.

Apart of the manuscript, we have collected all the information, certificates and additional files required in the publication rules

I am looking forward a positive answer regarding the acceptance for publication of this scientific contribution in World Journal of Gastroenterology.

Thanking in advance your attention, I am looking forward your news.

Yours sincerely

O. López Albors
albors@um.es
Full professor of Comparative Gross Anatomy
University of Murcia, Spain.